Author’s response to reviews

Title: Co-mutations of TP53 and KRAS Serve as Potential Biomarkers for Immune Checkpoint Blockade in Squamous-Cell Non-small Cell Lung Cancer: A Case Report

Authors:

Cheng Fang (jsycqnfangcheng@126.com)
Chu Zhang (zc.1988.ok@163.com)
Wei-Qing Zhao (zwq7860@163.com)
Wen-Wei Hu (huwenwei1979@163.com)
Jun Wu (wujun68@sina.com)
Mei Ji (jimei_90@126.com)

Version: 2 Date: 31 Aug 2019

Author’s response to reviews:

August 31, 2019

Dear editor,

Thank you for giving us the opportunity to improve my manuscript further. We have carefully read the comments of reviewers and made the revisions according to every reviewer’s comments. The revised portions were highlighted in red in the text.

To reviewer #1:

1. I suggest the authors add the response to comment #4 (in my previous report) to the manuscript.

Response: Thank you very much for your suggestion. The response to comment #4 has been added to the discussion section of manuscript and highlighted in red color in the revised manuscript.